Author's response to reviews

Title: Improving the implementation of perioperative safety guidelines using a multifaceted intervention approach: protocol of the IMPROVE study, a stepped wedge cluster randomized trial

Authors:

Yvette E.J.J.M. Emond (Yvette.Emond@radboudumc.nl)
Hiske Calsbeek (H.Calsbeek@radboudumc.nl)
Steven Teerenstra (Steven.Teerenstra@radboudumc.nl)
Gerrit J.A. Bloo (Gerrit.Bloo@radboudumc.nl)
Gert P. Westert (Gert.Westert@radboudumc.nl)
Johan Damen (Johan.Damen@radboudumc.nl)
André P. Wolff (Andre.Wolff@radboudumc.nl)
Hub C. Wollersheim (Hub.Wollersheim@radboudumc.nl)

Version: 6 Date: 17 December 2014

Author's response to reviews: see over
Editors, Implementation Science

17 December 2014
Cover letter
Article resubmission with requested changes

Dear professor Sales and professor Wensing,

Please find enclosed our manuscript entitled: 'Improving the implementation of perioperative safety guidelines using a multifaceted intervention approach: protocol of the IMPROVE study, a stepped wedge cluster randomized trial', which we have updated according to the CONSORT recommendations for cluster trials, and the CONSORT checklist for your consideration for publication as a study protocol in Implementation Science.

We look forward to your kind consideration of the manuscript.

Yours sincerely,
on behalf of all the authors,

Yvette E.J.J.M. Emond, MSc
Hub C. Wollersheim, MD, PhD